Author's response to reviews

Title: The association between urinary kidney injury molecule 1 and urinary cadmium in elderly during long-term, low-dose cadmium exposure: a pilot study

Authors:

Valérie Pennemans (valerie.pennemans@uhasselt.be)
Liesbeth M De Winter (liesbeth.dewinter@uhasselt.be)
Elke Munters (elke.munters@uhasselt.be)
Tim S Nawrot (tim.nawrot@uhasselt.be)
Emmy Van Kerkhove (emmy.vankerkhove@uhasselt.be)
Jean-Michel Rigo (jeanmichel.rigo@uhasselt.be)
Carmen Reynders (carmen.reynders@zol.be)
Harrie Dewitte (harrie.dewitte@gvhv.be)
Robert Carleer (robert.carleer@uhasselt.be)
Joris Penders (joris.penders@zol.be)
Quirine Swennen (quirine.swennen@uhasselt.be)

Version: 7 Date: 30 August 2011

Author's response to reviews:

in the abstract: in the results section the comma in the end should be replaced by a full stop
in table 3: "microalbumin" should be at the same position as the other biomarkers.